Author’s response to reviews

Title: The acute effects of alcohol on auditory thresholds

Authors:

T Upile (mrtupile@yahoo.com)
F Sipaul (Fabian@tiscali.com)
W Jerjes (waseem_wk1@yahoo.co.uk)
S Singh (s154singh@yahoo.com)
Sar Nouraei (Sarnouraei@hotmail.com)
M El Maaytah (elmaaytah@hotmail.com)
P Andrews (Peterandrews1625@sky.com)
J Graham (James14357@yahoo.com)
C Hopper (c.hopper@ucl.ac.uk)
A Wright (anthonywright@yahoo.co.uk)

Version: 4 Date: 4 September 2007

Author’s response to reviews:

Dear Editor

The requested changes were done

Kind regards

The authors